Craig Chambers CSE 505 (Fall ’95) Reading Assignments

Reading Assignment, Weeks #10 and #11

Using Types and Inheritance in OO Languages:

read about different ways to use and misuse inheritance

A Semantics of Multiple Inheritance:

a basic type-theoretic model of objects and subtyping;

look in particular at rules for when one value is a subtype of another value, especially for functions (i.e. the dreaded contravariance rule)

Intro to Trellis/Owl:

a nicely-designed statically-typed purely OO language; notice its type system and paramterization mechanism in particular; notice where Trellis uses the contravariant rule; it accesses instance variables by message sends!

A Proposal for Making Eiffel Type-Safe:

Eiffel is like Trellis except it uses a covariant rule;

see how this can lead to problems and how the type hole can be plugged

Inheritance is Not Subtyping

a formal study of fancier parameterization mechanisms, including bounded and F-bounded quantification

Cecil Language Spec

a description of the dynamically-typed core of Cecil; focus on objects, multi-methods, and fields

Typechecking and Modules for Multi-Methods a type system and a module system for Cecil; can ignore section on typechecking algorithm